

Citation: Mushi VP, Bhwana D, Massawe IS, Makunde W, Sebukoto H, Ngasa W, et al. (2024) Prevalence of onchocerciasis and epilepsy in a Tanzanian region after a prolonged communitydirected treatment with ivermectin. PLoS Negl Trop Dis 18(9): e0012470. https://doi.org/10.1371/ journal.pntd.0012470

Editor: Sung-Jong Hong, Chung-Ang University, REPUBLIC OF KOREA

Received: February 6, 2024

Accepted: August 18, 2024

Published: September 6, 2024

Copyright: © 2024 Mushi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files.

Funding: This research was commissioned by the National Institute for Health Research (grant number NIHR200134 to AS and CRN) using Official Development Assistance (ODA) funding. The grant compensated all authors for their time spent on the project activities. The funders had no role in study design, data collection and analysis,

RESEARCH ARTICLE

Prevalence of onchocerciasis and epilepsy in a Tanzanian region after a prolonged community-directed treatment with ivermectin

Vivian P. Mushi^{1,2}, Dan Bhwana³, Isolide S. Massawe³, Williams Makunde³, Hillary Sebukoto³, Willison Ngasa⁴, Joel Sengerema⁵, Athanas Mhina³, Paul M. Hayuma³, Henrika Kimambo⁶, Winifrida Kidima², William Matuja⁷, Josemir W. Sander^{8,9,10}, Helen Cross¹¹, Arjune Sen¹², Robert Colebunders¹³, Charles R. Newton^{12,14,15}, Bruno P. Mmbando^{3*}, EPInA Study Group¹

 Department of Parasitology and Medical Entomology, School of Public Health and Social Sciences, Muhimbili University of Health and Allied Sciences, Dar es Salaam, Tanzania, 2 Department of Zoology and Wildlife Conservation, College of Natural and Applied Sciences, University of Dar es Salaam, Dar es Salaam, Tanzania, 3 Tanga Research Centre, National Institute for Medical Research, Tanga, Tanzania, 4 Tanga Regional Referral Hospital, Tanga, Tanzania, 5 Bumbuli District Council, Tanga, Tanzania, 6 Department of Internal Medicine, Muhimbili National Hospital, Dar es Salaam, Tanzania, 7 Department of Neurology, Muhimbili University of Health and Allied Sciences, Dar es Salaam, Tanzania, 8 Department of Clinical and Experimental Epilepsy, UCL Queen Square Institute of Neurology, London WC1N 3BG & Chalfont Centre for Epilepsy, London, United Kingdom, 9 Neurology Department, West China Hospital, Sichuan University, Chengdu, China, 10 Stichting Epilepsie Instellingen Nederland (SEIN), Heemstede, The Netherlands, 11 Developmental Neurosciences Research and Teaching Department, UCL Great Ormond Street Institute of Child Health, London, United Kingdom, 12 Oxford Epilepsy Research Group, Nuffield Department of Clinical Neurosciences, University of Oxford, Oxford, United Kingdom, 13 Global Health Institute, University of Antwerp, Antwerp, Belgium, 14 Neuroscience Unit, KEMRI-Wellcome Trust Research Programme, Kilifi, Kenya, 15 Department of Psychiatry, University of Oxford, Oxford, United Kingdom

¶ Membership of EPInA Study Group is provided in Supporting Information file <u>S1 Acknowledgments</u>.
* bruno.mmbando@nimr.or.tz

Abstract

Introduction

Epidemiological evidence suggests that *Onchocerca volvulus* is associated with epilepsy, although the exact pathophysiological mechanism is unknown. Mahenge is an endemic focus of onchocerciasis, with the longest-running ivermectin treatment intervention in Tanzania. We assessed the prevalence of onchocerciasis and epilepsy after 25 years of control using ivermectin.

Methods

This was a population-based cross-sectional study in 34 villages in Mahenge in 2021. Community health workers conducted door-to-door household surveys to enumerate the population and screen for individuals suspected of epilepsy using a standardised questionnaire. Trained physicians confirmed epilepsy. Children aged 6–11 years were screened for onchocerciasis antibodies using the Ov16 rapid test. Villages were stratified into three altitude levels (low [<400], medium [400–950], and high [>950 meters above sea level]) as a proxy for decision to publish, or preparation of the manuscript. The views expressed in this publication are those of the author(s) and not necessarily those of the NHS, the National Institute for Health Research, or the Department of Health and Social Care.

Competing interests: The authors have declared that no competing interests exist.

rapids, which black flies favour for breeding sites. Incidence of epilepsy was estimated as a ratio of new cases in the year preceding the survey per 100,000 population.

Results

56,604 individuals (median age 20.2 years, 51.1% females) were surveyed. Onchocerciasis prevalence in children was 11.8% and was highest in villages at medium (21.7%) and lowest in low altitudes (3.2%), p<0.001. Self-reported use of ivermectin was 88.4%. Epilepsy prevalence was 21.1 (95%CI: 19.9–22.3) cases per 1000 persons and was highest in medium (29.5%) and lowest in the lowlands (12.7%). The odds ratio (OR) of having epilepsy was significantly higher in females (OR = 1.22, 95%CI: 1.08–1.38), middle altitudes (OR = 2.34 [95%CI: 2.04–2.68]), and in individuals positive for OV16 (OR = 1.98 [95%CI:1.57–2.50]). The incidence of epilepsy a year before the survey was 117 (95%CI: 99.7–160.4) cases per 100,000 person-years.

Conclusion

Despite ivermectin use for 25 years, the prevalence of onchocerciasis and epilepsy remains high. It is crucial to strengthen bi-annual ivermectin treatment and initiate interventions targeting vectors to control onchocerciasis and epilepsy in the area.

Author summary

Onchocerciasis is endemic in Mahenge, with epidemiological evidence that the parasite (*Onchocerca volvulus*) may be associated with epilepsy. The high burden of the disease in Mahenge led to the introduction of Community Directed Treatment with Ivermectin (CDTI) in 1997. Despite the long history of CDTI implementation in Mahenge, the persistent transmission of the parasite is still reported. A population-based cross-sectional study was conducted to establish the current prevalence of onchocerciasis and epilepsy 25 years after the start of the CDTI programme.

The prevalence of epilepsy was 21.1 (95%CI: 19.9–22.3) cases per 1000 people. The incidence of epilepsy one year preceding the survey was 117 (95%CI: 99.7–160.4) per 100,000 person-years. Despite high self-reported coverage of ivermectin (88%), active onchocerciasis transmission was observed among children aged 6–11 years, and more than half of the individuals with epilepsy (54.1%) were serologically positive for onchocerciasis. The predictors for epilepsy were sex (female), age \geq 10 years, being a resident of Mahenge for \geq 20 years, especially in villages located at a medium altitude, and being onchocerciasis positive. Strengthening of processes to ensure bi-annual ivermectin treatment is recommended.

Introduction

Onchocerciasis, a parasitic disease due to *Onchocerca volvulus* (*O. volvulus*), causes significant morbidity and mortality in endemic areas [1]. The disease vectors are infected black flies (*Simulium* spp), which reproduce in fast-flowing water, thus this disease is called river blindness [2]. Onchocerciasis is responsible for skin (lesions, onchodermatitis) and ocular (conjunctivitis, sclerosing, keratitis, and blindness) manifestations [3]. Over 200 million people are

at risk of onchocerciasis, with almost all cases occurring in sub-Saharan Africa. It is estimated that more than 40 million people are currently infected [2,4].

The high disease burden led to the introducing of control interventions such as community-directed treatment with ivermectin (CDTI) and vector control with environmentally safe methods [2]. CDTI is a core strategy recommended by the WHO and was predicted to control the infestation within 16–18 years at coverage of at least 80% of the population [5]. In some endemic areas, CDTI deployment has significantly reduced onchocerciasis transmission [6,7]. In other regions, despite decades of CDTI, there are still persistent onchocerciasis transmission [8,9] and onchocerciasis-associated morbidities such as epilepsy [10–12].

A high prevalence of epilepsy has been reported in onchocerciasis endemic areas. Recent epidemiological evidence from sub-Saharan Africa suggests that *O. volvulus* is associated with epilepsy, and is referred to as onchocerciasis-associated epilepsy (OAE) [11–15]. Some suggest that nodding syndrome, characterised by head nodding attacks, cognitive deterioration, and stunted growth, coupled with other presentations such as Nakalanga syndrome (features include delayed puberty, mental impairment, and stunted growth), are forms of OAE [16]. The exact pathophysiological mechanism by which *O. volvulus* is associated with epilepsy remains unknown.

In Tanzania, onchocerciasis is endemic, with more than 6.5 million people at risk [17]. The prevalence of onchocerciasis varies across the country, with the Mahenge area having the highest regional prevalence since the 1960s [17]. The annual CDTI programme in Tanzania started in Mahenge in 1997 [12]. Two decades later, onchocerciasis transmission was still active, with some areas reporting a prevalence of high seropositivity, up to 40% amongst the rural population [11]. Low CDTI uptake and adherence, suboptimal programme implementation, and other challenges are considered responsible for this high prevalence [18]. Persistent transmission is regarded as the main reason for the continued high incidence and prevalence of epilepsy in Mahenge [10,11,19].

An OAE incidence rate of 131 cases per 100,000 person-years was reported in four villages (two peri-urban and two rural villages) in 2018, with rural areas having the highest rates [11]. As a result, in 2019, the Tanzania Neglected Tropical Diseases Control Programme (TNTDCP) initiated bi-annual ivermectin treatment in Mahenge. Current data on the burden of onchocerciasis and epilepsy in Mahenge following this supervised administration of ivermectin is limited. We assessed the current burden of onchocerciasis and epilepsy in the Mahenge area following the long-term use of ivermectin (25 years) for onchocerciasis control. A principal aim was to provide national agencies with an update on the current status of onchocerciasis transmission and associated morbidities. This will enable data-driven programmatic changes to accelerate onchocerciasis control and help the country to eliminate onchocerciasis.

Materials and methods

Ethics statement

Ethical approval was provided by the National Institute of Medical Research Ethics Committee (NIMR/HQ/R.8a/Vol.IX/3582). Permission to conduct the study in the Mahenge area was obtained from the appropriate administrative units. All participants were given information about the study before being asked to provide written consent. Only consenting individuals were enrolled. Adults (\geq 18 years) provided their informed consent by signing the informed consent. For children and minors, permission was sought from their parents or guardians, who signed the consent forms. Additionally, children aged 12–17 years were asked for verbal assent to be involved in the study.

Study area and demographics

This study was carried out in Ulanga district, one of the seven districts of the Morogoro region. The district lies at latitude 8° 40' 59" S and longitude 36° 43' 0" E and is bordered to the east by the Lindi region, to the west and north by the Kilombero district, and south by the Ruvuma Region. The district has an area of 24,460 km² divided into four divisions, 21 wards, and 59 administrative villages with an approximate population of 265,000 [20]. Thirty-four villages ranging from 306 to 1200 meters above sea level (mASL) were involved.

The district experiences an annual rainfall of 1200mm to 1800mm, and temperatures range from 22°C to 33°C. The Mahenge area is mountainous, with more than 40 fast-flowing rivers and streams providing breeding habitats and supporting *Simulium damnosum s.I* survival (vector in the district).

Study design and population

A population-based cross-sectional study was conducted between August and November 2021. All households and household members living in selected villages were eligible to participate.

Selection of the study villages

Initially, a brief health facility-based survey was implemented in health facilities in the Malinyi and Ulanga districts to assess hospital attendance. Implemented between May and June 2021, this survey aimed to estimate the number of people with epilepsy in the catchment area. A list containing names of wards, villages, and population projections for 2018, 2019, and 2020 was obtained from the district executive office in the Ulanga district. Village visits were organised concurrently to collect information from village executive officers to complement the health facility data. Community health workers (CHWs) in villages provided information on the number of people with epilepsy as they are more conversant with health information in their settings. Where data was missing, CHWs were requested to liaise with community members to ascertain the number of people with epilepsy. The prevalence of epilepsy was estimated as the number of epilepsy cases obtained from health facilities or village executive officers per 1000 population. Villages with a high estimated prevalence of epilepsy were selected purposively to be involved.

Epilepsy screening and assessment of ivermectin coverage

A two-step approach was used to identify people with epilepsy. A five-question validated epilepsy screening questionnaire was used in the door-to-door household survey [21]. Trained CHWs embedded in the study villages administered the questionnaire to all individuals in a household (S1 Text). Those with at least one positive answer to the questions (screen positive) were invited to a central point for clinical evaluation and diagnostic confirmation by a trained physician (DB, WM, HS, WN, and JS) (S2 Text). Individuals diagnosed with epilepsy were also tested for *O. volvulus* antibody detection using the Ov16 rapid test. People who failed to attend the central point were traced back to their households for diagnostic confirmation. Those newly confirmed to have epilepsy were referred to a nearby health facility for care. Participants were interviewed about their history of ivermectin use. Answers from children were confirmed by their parents or legal guardians.

Determining onchocerciasis active transmission

The WHO recommends testing children aged 6–10 years to monitor the active transmission of *O. volvulus* in the community under CDTI. The test identifies the existence of active

transmission in the community but cannot differentiate between past exposure and active transmission. In children aged 6–11, a finger prick blood sample was obtained for *O. volvulus* antibody detection using the Ov16 antigen rapid test (Standard Diagnostics, Incorporated, Gyeonggi-do, Republic of Korea).

Outcome and independent variables

The study outcomes were the proportion of people who tested positive for onchocerciasis using *O. volvulus* IgG4 antibodies detected by Ov16 rapid tests and those confirmed to have epilepsy according to accepted criteria. Independent variables included sex (female or male), age groups, residential village, duration of residence in the study area, altitude (306–1200 meters above sea level), family size, and history of recent use of ivermectin.

Data management and analysis

A database was created and uploaded using the open-source software Open Data Kit (ODK) [https://opendatakit.org/]. Data were collected using tablets and uploaded daily to a server, where the data was checked for consistency. CHWs resolved queries raised. The ODK data was exported to STATA version 17 (Stata Corp Inc., TX, USA) and R. version 4.0.3 (R Core Team 2020, Vienna, Austria) for analysis. Descriptive statistics were used to summarise the categorical variables into counts/frequency, while continuous variables, median (with inter-quartile range [IQR]), and/or means with standard deviations were used. The onchocerciasis seroprevalence, prevalence of epilepsy, and ivermectin uptake were further summarised based on the villages and by altitude. To reduce the bias in underreporting epilepsy due to non-response for various reasons, we assumed a proportion of confirmation to be similar between those who responded and the non-responders. Participants were stratified into different age categories. Altitude was grouped into three levels (low, medium, and high) as a proxy for rapid water flows, favouring breeding sites for *Simulium* spp.

The incidence of epilepsy was estimated as the number of epilepsy cases in which seizures started in the previous year, divided by the population size for the period. A one-year period is likely to have less recall bias but may be affected by underreporting due to stigma. We also estimated the incidence based on the five years preceding the study, which is more robust in estimating uncertainty around the mean due to more data in the numerator. We assumed a uniform population across the five years.

Proportions were compared using a χ^2 -test, while means were compared using a t-test. Univariate logistic regression was used to assess the association between explanatory variables and epilepsy, whereas a multivariate model was used to control for confounding variables. All independent variables with p-value <0.1 were included in the multivariate model to adjust for the confounding variables. The findings were reported as adjusted odds ratios (aOR), and a p-value < 0.05 was deemed significant.

Definitions of key terms

Epilepsy cases were defined and confirmed according to the older classification of ILAE as the presence of two or more unprovoked seizures more than 24 hours apart due to the unavailability of brain imaging or electroencephalography [22]. This allows for meaningful comparisons with previous results in Mahenge and minimises the variability introduced by changes in diagnostic criteria. A household was defined as a person or people living together and eating from the same cooking pot [11]. An onchocerciasis-positive case was defined as someone who tested positive for *O. volvulus* IgG4 antibodies using Ov16 tests.

Results

Baseline characteristics of the study participants

During the house-to-house survey, 14,644 households were visited comprising 56,604 individuals. More than half of the participants were females (28,895, 51.1%), with the median age of participants being 20.2 years (inter-quartile range (IQR): 9.3–39.1) (Table 1).

The study area included villages with median altitudes ranging from 306 to 1086 meters above sea level. Magereza village had the highest population (3,896 people), while Ikungua village had the lowest (508 people), Fig 1.

Prevalence of epilepsy among the study participants

Fig 2 displays the prevalence of epilepsy and Ov16 test findings by village and altitude in the study population. A total of 1946 (3.4%) individuals were suspected of having epilepsy, of whom 155 (8.0%) were not available for confirmation of epilepsy while 1194 were confirmed to have epilepsy, giving an overall prevalence of 21.1 (95%CI: 19.9–22.3) cases per 1000 persons. Adjust for participants who did not show up for confirmation, assuming a similar proportion of confirmation (61 for 100 suspected), and the prevalence of epilepsy was 22.8 cases per 1000 (95%CI, 21.6–24.0).

Based on confirmed epilepsy, the villages between medium altitudes of 400–950 meters above sea level had a significantly higher prevalence of 3.1% (95%CI: 2.9–3.3) compared to villages at low and high altitudes (Table 1). Isaka village, at a medium altitude, had the highest prevalence (4.8%), while villages with the lowest prevalence in the low and high altitudes were Kichangani (1.0%) and Epanko (0.8%; Fig 2).

Characteristic	Population size	Number without epilepsy	Number with epilepsy, n (%)	Test statics, P-value	
Number of households, n	14,644	13,581 (92.7)	1063 (7.3)		
Population	56,604	55,463 (98.0)	1141 (2.0)		
Age (years), median (IQR)	20.2 (9.3–39.1)	20.1 (9.3–39.1)	27.8 (17.2–39.0)	$\chi^2 = 183,$ p< 0.001	
Sex					
Males (%)	27,709 (48.9)	27,173 (98.1)	536 (1.9)	$\chi^2 = 8.1,$	
Females (%)	28,895 (51.1)	28,237 (97.7)	658 (2.3)	p = 0.005	
Age group (years), n (%)					
0-9	14,935	14,786 (99.0)	149 (1.0)	$\begin{array}{c} \chi^2_{(6)} = 290, \\ p < 0.001 \end{array}$	
10-19	13,079	12,868 (98.4)	211 (1.6)		
20-29	8,192	7,891 (96.3)	301 (3.7)		
30-39	6,753	6,502 (96.3)	251 (3.7)		
40-49	5,352	5,229 (97.7)	123 (2.3)		
50-59	3,734	3,662 (98.1)	72 (1.9)		
60+	4,471	4,388 (98.1)	83 (1.9)		
Missing	88	84 (95.4)	4 (4.6)		
Altitude (meters above sea level)				
Low (<400m)	20,860	20,527(98.4)	333 (1.6)	$\chi^2_{(2)} = 159,$ p< 0.001	
Medium (400-950m)	21,680	21,010 (96.9)	670 (3.1)		
High (>950m)	14,064	13,873 (98.6)	188 (1.4)		
Ivermectin use in Aug 2018 for	\geq 5 years				
Yes (%)	42,491 (88.4)	41,558 (88.4)	933 (86.9)	$\chi^2 = 2.4,$ p = 0.123	
No (%)	5,596 (11.6)	5,455 (11.6)	141 (13.1)		

Table 1. Demographic characteristics of the study population by confirmed epilepsy status.

https://doi.org/10.1371/journal.pntd.0012470.t001



Fig 1. Summary of study villages' population and altitude. The bars indicate the enumerated study population size, while the dashed line shows the median altitude of the study villages.

https://doi.org/10.1371/journal.pntd.0012470.g001

The prevalence of epilepsy was significantly higher in females (2.3, 95%CI: 2.1-2.5%) than in males (1.9%, 95%CI: 1.8-2.1), p = 0.005. In addition, a more significant epilepsy burden was seen in the age groups of 20–39 years (3.7%), with females aged 30–39 years having the highest prevalence (Fig 3).

Ov16 seroprevalence among individuals with epilepsy

The overall prevalence of Ov16 positivity among people with epilepsy was 54.1% (474/876) and was highest in villages located at medium altitudes (64.2%), followed by villages at high altitudes (49.6%) and lower laying villages (36.6%), χ^2 (3) = 52.2, p<0.001.

Ov16 seroprevalence in people with epilepsy increased with age in the entire study area. In the low and high altitudes, seropositivity rose from 0% and 15.4% in the group aged 0–9 years to a peak (>65%) in those aged 50–59. In middle altitudes, the prevalence increased from 25.5% in the age group 0–9 years to >65% in the age group 10–19 years, suggesting the risk of infection is higher in this altitude band (Fig 3A).

The prevalence of epilepsy increased with age, reaching its maximum in people aged 20–39 years. It was also observed that, except for 20–29 years, the prevalence was higher among females than males (Fig 3B).

Ov16 seroprevalence and risk factors among children aged 6-11 years

Ov16 seroprevalence amongst the children in this age range was 11.8% (n = 3731, 95%CI: 10.8–12.9), with a high prevalence observed among the children from Isaka village (57.4%). By contrast, none of the children from Chirombola were infected (Fig 2).

Ov16 seroprevalence was highest in the villages located at medium altitudes (21.7%), followed by highlands (9.3%), and lowest in the lowlands (3.2%; p<0.001). The logistic regression model shows the risk (odds ratio) of being Ov16 positive was 8.45 times (95%CI: 6.16–11.61,



Fig 2. Prevalence of confirmed epilepsy and onchocerciasis by village and altitude (low, medium, and high altitudes). Panel A shows the prevalence of epilepsy (red line), while the dotted line is the median altitude. Panel B shows the prevalence of Ov16 positivity (solid line), while the median altitude is plotted using a dotted line.

https://doi.org/10.1371/journal.pntd.0012470.g002



Fig 3. Prevalence of Ov16 positivity by age group (A) among participants with confirmed epilepsy by strata for low (full line), medium (dashed line), and high altitudes (dotted line). Panel B shows the prevalence of epilepsy by age group for males (black bars) and females (red bars).

https://doi.org/10.1371/journal.pntd.0012470.g003

Variable	Univariate analysis		Multivariate analysis	Multivariate analysis	
	OR (95%CI)	P-value	aOR (95%CI)	P-value	
Sex					
Males	1				
Females	0.95 (0.78-1.16)	0.615			
Age (years)	1.26 (1.17–1.35)	< 0.001	1.27 (1.18–1.36)	< 0.001	
Duration of residence (years)	1.12 (1.06–1.19)	< 0.001			
Family size (people)	1.04 (0.99-1.09)	0.160			
Ivermectin uptake					
Yes	1				
No	1.01 (0.67–1.50)	<0.972			
Altitude (meters above sea level)					
Low (<400m)	1		1		
Medium (400-950m)	8.35 (6.09-11.46)	< 0.001	8.45 (6.16–11.61)	<0.001	
High (>950m)	3.10 (2.13-4.52)	<0.001	3.17 (2.17-4.62)	<0.001	

Table 2.	Risk factors for Ov16	positivity among	g the study partici	pants (age 6–11 vears).
			g ente occasi parterer	

sOR: Crude Odds Ratios, aOR: Adjusted Odds Ratios (age, ivermectin uptake and altitude)_

https://doi.org/10.1371/journal.pntd.0012470.t002

p<0.001) in children living at medium altitudes. For those in the highlands, it was 3.17 (95% CI: 2.17–4.62, p<0.001) when compared to children living in the lowlands (Table 2).

Most children had received their first dosage of ivermectin by the age of six years, but more than 15 per cent of children of this age range from communities located at medium altitudes tested positive for Ov16, with the highest prevalence observed among children aged 11 years (**Fig 4**).

Self-reported uptake of ivermectin among the study participants

The self-reported uptake of ivermectin in the last round of CDTI was 88.4% (42,492/48,087) among the study participants aged \geq 5 years, with the highest uptake recorded in Chirombola (94.3%) and Isyaga (94.1%) villages. The lowest uptake recorded was in Magereza (67.8%), (S1 Table).

Among the participants aged ≥ 5 years and screened for onchocerciasis, those who did not take ivermectin in the last round had a higher prevalence of Ov16 compared to those who took ivermectin (24.4% vs. 20.0%, $\chi^2 = 4.4$, p = 0.037). There was no association between the history of use of ivermectin and epilepsy ($\chi^2 = 2.4$, p = 0.123), neither correlation between village coverage of the ivermectin and prevalence of epilepsy (Pearson correlation = 0.09, 95%CI: (-0.28–0.44), p = 0.636

Predictors for epilepsy among the study participants

<u>Table 3</u> illustrates results from univariate and multivariate models showing the association between epilepsy and explanatory variables. In the multivariate model, Ov16 status was omitted since less than 10% of the participants (majority 6–10 years) were screened for onchocerciasis; instead, reported the effect of the Ov16 test is based on a multivariate model fitted for 4464 records.

Estimates from univariate and multivariate models (49,992 records; $R^2 = 4\%$) were similar, with slight variations by age group. The low score indicates that only about 4% of the observed variations in the prevalence could be explained by variables included in the model. Factors significantly associated with developing epilepsy in the multivariate model were: sex, age (by



Fig 4. Prevalence of Ov16 positivity with associated 95%CI amongst children aged 6–11 years by altitude strata. The full line represents low altitudes; the dashed line represents middle altitudes; the dotted line represents high altitudes. The error bars are plotted to avoid overlay between the altitudes strata.

https://doi.org/10.1371/journal.pntd.0012470.g004

group), duration of residency, altitude, and onchocerciasis status (<u>Table 3</u>). The multivariate model (4464 records) suggests that people with positive Ov16 results had a risk of having epilepsy (aOR) of 1.98 (95%CI: 1.57–2.50), p<0.001).

Incidence of epilepsy in Mahenge

Sixty-six participants developed epilepsy in the year preceding the survey, giving an incidence of 117 (95%CI: 90–148) per 100,000 person-years. Residents of villages in the middle altitude had the highest incidence: 184 (95%CI: 131–251) cases per 100,000 person-years, 96 (95%CI: 59–148) in the lowland, while in the highlands it was 42 (95%CI: 17–93) cases per 100,000 person-years. If we adjust for participants who did not have confirmation, the epilepsy incidence was projected at 127 cases per 100,000 (95%CI, 99–160).

The five-year incidence was slightly lower than that based on one year: 96 (95%CI: 85–108) per 100,000 person-years. Table 4 shows factors associated with epilepsy incidence over the past five years. Compared to the lowlands, the incidence of epilepsy was 39% higher in the medium-altitude villages and 40% lower in the highlands. Incidence was highest in young children (individuals aged 10 to 19 years) and decreased with age, while females had a significantly higher incidence of 32% compared to males (p = 0.023). The trend of confirmed epilepsy incidence based on five years by altitude strata and age group is presented (Fig 5).

Discussion

We present epidemiological data on the current burden of onchocerciasis and epilepsy in Mahenge following 25 years of CDTI in this region. Previous studies in Mahenge reported changes in onchocerciasis and epilepsy burden, showing persistent transmission of

Variable	Univariate analysis		Multivariate analysis (n = 49,992)		
	OR (95%CI)	P-value	aOR (95%CI)	P-value	
Sex					
Males	1		1		
Females	1.18 (1.05–1.33)	0.005	1.22 (1.08–1.38)	0.002	
Age (years)					
0-9	1		1		
10-19	1.63 (1.32-2.01)	< 0.001	1.60 (1.28–1.99)	< 0.001	
20-29	3.79 (3.10-4.62)	< 0.001	2.30 (1.79–2.96)	<0.001	
30-39	3.83 (3.12-4.70)	< 0.001	2.27 (1.75-2.96)	<0.001	
40-49	2.33 (1.83-2.97)	< 0.001	1.42 (1.02–1.96)	0.036	
50-59	1.95 (1.47-2.59)	< 0.001	1.21 (0.83–1.76)	0.329	
60+	1.88 (1.43-2.46)	< 0.001	1.20 (0.82–1.75)	0.352	
Duration of residence (years)					
0–19	1		1		
20-39	2.89 (2.53-3.29)	< 0.001	1.91 (1.57–2.32)	<0.001	
40+	1.46 (1.23–1.74)	< 0.001	1.59 (1.17-2.16)	0.003	
Family size (people)					
≤5	1				
6-11	0.80 (0.70-0.90)	< 0.001			
≥12	1.11 (0.68–1.84)	0.671			
Ivermectin uptake (\geq 5 years)					
Yes	1				
No	1.16 (0.96–1.41)	0.113			
Altitude (mASL)					
Low (<400m)	1		1		
Medium (400-950m)	1.97 (1.72–2.24)	< 0.001	2.34 (2.04–2.68)	< 0.001	
High (>950m)	0.85 (0.71-1.02)	0.001	0.82 (0.68-0.98)	0.029	
*OV16 tests status					
Negative	1		1		
Positive	6.81 (5.81-7.98)	< 0.001	1.98 (1.57-2.50)	< 0.001	

Table 3. Predictors of epilepsy among the study participants.

OR: Crude Odds Ratios, aOR: Adjusted Odds Ratios, *Based on a multivariate model with 4,464 records and pseudo R² = 37.4%

https://doi.org/10.1371/journal.pntd.0012470.t003

onchocerciasis and high prevalence and incidence of epilepsy despite two and half decades of CDTI [8,10,11,19,23–25]. Our findings are similar to those of the multicentric assessment of the seroprevalence of *O. volvulus* among people with epilepsy in onchocerciasis foci in the Democratic Republic of Congo, Cameroon, Tanzania (Mahenge), and Uganda, which reported prevalence ranging from 35.2% to 59.7% [15]. The observed prevalence of epilepsy and Ov16 among people with epilepsy in this study adds epidemiological evidence to the hypothesis that *O. volvulus* is associated with epilepsy, even though the pathophysiological mechanisms triggering seizures are still not well understood [16]. There are different hypotheses about the possible causes of the condition, including invasion of microfilariae into the central nervous system [26], deposition of pathogenic tau protein in the brain leading to neuronal dysfunction and neurodegenerative disease [27], and the cross-reactivity of Leomodin-1 antibodies with *O. volvulus* protein [28]. The evidence for any of these hypotheses is inconsistent between studies. In our study, the incidence based on the previous year was similar to the

Variable	IRR	95%CI	P-value
Altitude (mASL)			
Low (<400m)	1		
Medium (400-950m)	1.39	1.06-182	0.017
High (>950m)	0.60	0.41-0.88	0.009
Age (years)			
0-9	1		
10-19	1.05	0.79-1.39	0.752
20-29	0.69	0.47-1.01	0.055
30-39	0.47	0.30-0.75	0.001
40-49	0.23	0.11-0.46	<0.001
50–59	0.32	0.15-0.65	0.002
60+	0.23	0.11-0.50	<0.001
Sex Males	1	-	
Females	1.32	1.04-1.69	0.023
Cons (per 100,000)	83.0	52.2-131.9	<0.001

Table 4. Results from multivariable Poisson regression model showing incidence rate ratios (IRR).

Cons: estimates baseline incidence rate, IRR: Incidence rate

https://doi.org/10.1371/journal.pntd.0012470.t004

earlier reported figures in four villages in Mahenge in 2017/2018 [11]. Five-year incidence, also previously used, was slightly lower, potentially owing to more villages from varying altitudes with more intense onchocerciasis transmission compared to the 2017/2018 study that





https://doi.org/10.1371/journal.pntd.0012470.g005

used data from only four villages [11]. The incidence of epilepsy in our study remains high compared to reports from other sub-Saharan countries such as Ghana, Ethiopia, and South Africa [29–31]. The epilepsy prevalence in Mahenge correlated with onchocerciasis exposure, with the villages located at medium altitudes having a high prevalence of onchocerciasis and epilepsy. The risk of acquiring onchocerciasis was eight times higher for the children residing in the villages at the medium altitudes than those at the lower and higher altitudes. An explanation for this could be that as rivers start to flow at higher altitudes, it is at medium altitudes that rapids occur, which are rich in oxygen and favour the breeding and survival of the *Simulium*. Hence, the high risk of transmission is in villages located at medium altitudes. This finding provides essential information about specific communities that may need additional control measures, particularly complemented vector control interventions [32], to accelerate the control of onchocerciasis and reduce epilepsy incidence.

The high prevalence of epilepsy in Ebuyu village may be due to variability of the altitude and environmental conditions in the western part of Ebuyu, which Isyaga village shares. Data from Ebuyu village shows that the prevalence of epilepsy in households located at altitudes \geq 400 meters above sea level was 5.57%, while for those at lower altitudes, the prevalence was 2.8% (S1 Table). Conversely, the high prevalence in Ikungua village could be due to its location near the Luli River, which provides a conducive environment for *Simulium* to hide. The low Ov16 positivity rate in these villages could be due to low exposure to infective bites.

The observed higher prevalence of epilepsy in females may be related to activities that expose them to fast-flowing rivers or streams where they can be bitten by *Simulium* (for example, washing clothes or utensils). Additionally, male migration to other regions in search of employment, leaving females in the villages for extended periods, may account for the difference in prevalence between males and females, especially in the age group 30–39 years. Our findings differ from the study conducted in four villages of Mahenge in 2017, which showed no difference in the prevalence of epilepsy between males and females [11]. According to age categories, those aged 20 to 39 had the highest prevalence of epilepsy, possibly as a result of a shift in peak prevalence from the young individuals as a result of several years of CDTI implementation, in line with the results of other studies [11,33].

The overall prevalence of Ov16 in children aged 6–11 years was 11.8%, similar to the prevalences observed before the introduction of bi-annual CDTI in the area in 2019. This was to be expected, as most children were already infected before the switch to bi-annual CDTI. Switching from an annual to a biannual CDTI strategy reduced the incidence of epilepsy in four rural villages included in this study where this was assessed [25]. A similar finding was observed in Uganda and South Sudan onchocerciasis endemic areas [34,35]. In 2021, in Mahenge, the selfreported uptake of ivermectin was high, presumably because of OAE awareness campaigns by the NTD programme and the bi-annual distribution of ivermectin. In 2022, a study among school children in the Mahenge area in Mahenge B, Makanga, and Msogezi villages however, found that about half of the children eligible for ivermectin did not take the drug during the last round of MDA, which suggests over-reporting of ivermectin intake in our study, especially in households where there was only one person providing information for the entire household [24]. Sub-optimal uptake has been observed in some villages, which may impact the success of the onchocerciasis interruption and lengthen the time that ivermectin is distributed, and in turn, contribute to the persistence of onchocerciasis-associated epilepsy (S1 Table).

There was a high prevalence of Ov16 positivity among the people with epilepsy who did not receive ivermectin in the last distribution round compared to those who took it. Ivermectin, administered once or twice a year, has been shown to reduce the microfilaria load, relieve extreme itching, and prevent the progression toward blindness [23,36]. According to studies

from other onchocerciasis-endemic countries, groups with low ivermectin uptake serve as a reservoir of infection and could develop morbidities associated with onchocerciasis, such as epilepsy [37–39].

Factors that increased the odds of developing epilepsy in Mahenge were being female; being older than ten years, being a resident of Mahenge for \geq 20 years, especially in villages located at medium altitude, and being onchocerciasis positive. Being infected with onchocerciasis with a high microfilariae load could directly or indirectly trigger epilepsy. Similarly, being a resident of Mahenge for \geq 20 years increases the risk of exposure to *Simulium* spp, resulting in onchocerciasis, which may later on trigger epilepsy. The age group of 20–39 years had higher odds of having epilepsy in Mahenge than other age groups. This finding is similar to other studies conducted in onchocerciasis endemic areas [13,40]. Studies have reported that before the introduction of CDTI, the risk of developing epilepsy was among children aged 10– 20 years, but after CDTI initiation, a shift of the highest prevalence of epilepsy was observed in the 20–39 years age group [11,13,40].

Study limitations

Our study has certain limitations. *O. volvulus* IgG4 antibodies were detected in the laboratory using the Ov16 rapid diagnostic test rather than the more sensitive Ov16 ELISA, which could impact the assessment of onchocerciasis seroprevalence. The use of the Ov16 ELISA could have resulted in more positive cases.

It should be noted that the bi-annual CDTI was only introduced three years before the survey. As a result, all children between the ages of 6–10 who tested positive for Ov16 had already been infected before the implementation of the bi-annual CDTI. Another limitation is that no entomological investigations were performed to identify blackfly breeding sites and infection rates.

Disclosure of epilepsy is a challenge in many communities, including Mahenge, owing to stigma [41], which may result in underreporting. People with epilepsy are at increased risk of dying due to seizure-related adverse events such as burns, status epilepticus, and drowning, which may affect the incidence rate reported.

The self-reported uptake of ivermectin could be subjected to recall bias because it requires recalling previous information, which could cause an over/underestimation of the prevalence of ivermectin uptake. The two-step approach used to identify people with epilepsy involved many interviewers, making it potentially susceptible to interviewer bias. This was, though, minimised by training the interviewers before and during data collection.

Recommendations

The prevalence of onchocerciasis and epilepsy remains high in Mahenge despite using ivermectin for 25 years. We recommend strengthening the ongoing bi-annual ivermectin treatment, especially in the villages with a high prevalence of onchocerciasis and epilepsy. Formulating an appropriate ivermectin dose is needed for children <5 years since they acquire onchocerciasis before the recommended age for ivermectin uptake. This is an opportune time to introduce vector (*Simulium*) control interventions, especially in the villages 400–950 meters above sea level, to accelerate the fight against onchocerciasis further and prevent the potential future development of epilepsy.

Supporting information

S1 Text. Household Screening Questionnaire. (DOCX)

S2 Text. Neurological Questionnaire. (DOCX)

S1 Acknowledgments. Members of the EPInA Study Group. (DOCX)

S1 Table. Table shows the study villages and distribution of median altitude, ivermectin coverage, and prevalence of Ov16 results and epilepsy. (DOCX)

S1 Data. The data set used for analysis. (XLSX)

Acknowledgments

The authors would like to thank the community members of Mahenge for participating in this study, community leaders for mobilising the study participants, and data collectors for their assistance in data collection. Special thanks should go to the Epilepsy Pathway Innovation in Africa (EPInA) project and their team members for support in accomplishing this work.

Author Contributions

Conceptualization: Vivian P. Mushi, Dan Bhwana, Willison Ngasa, William Matuja, Josemir W. Sander, Helen Cross, Arjune Sen, Charles R. Newton, Bruno P. Mmbando.

Data curation: Bruno P. Mmbando.

Formal analysis: Vivian P. Mushi, Bruno P. Mmbando.

Funding acquisition: Arjune Sen, Charles R. Newton.

- **Investigation:** Vivian P. Mushi, Dan Bhwana, Isolide S. Massawe, Williams Makunde, Hillary Sebukoto, Willison Ngasa, Joel Sengerema, Athanas Mhina, Paul M. Hayuma, Henrika Kimambo, Winifrida Kidima, William Matuja, Josemir W. Sander, Helen Cross, Arjune Sen, Robert Colebunders, Charles R. Newton, Bruno P. Mmbando.
- Methodology: Vivian P. Mushi, Dan Bhwana, William Matuja, Josemir W. Sander, Arjune Sen, Robert Colebunders, Charles R. Newton, Bruno P. Mmbando.

Project administration: Bruno P. Mmbando.

Software: Bruno P. Mmbando.

Supervision: Bruno P. Mmbando.

Validation: Bruno P. Mmbando.

Writing - original draft: Vivian P. Mushi, Dan Bhwana, Bruno P. Mmbando.

Writing – review & editing: Vivian P. Mushi, Dan Bhwana, Isolide S. Massawe, Williams Makunde, Hillary Sebukoto, Willison Ngasa, Joel Sengerema, Athanas Mhina, Paul M. Hayuma, Henrika Kimambo, Winifrida Kidima, William Matuja, Josemir W. Sander, Helen Cross, Arjune Sen, Robert Colebunders, Charles R. Newton, Bruno P. Mmbando.

References

1. Crump A, Morel CM, Omura S. The onchocerciasis chronicle: from the beginning to the end? Trends Parasitol. 2012; 28(7):280–8. https://doi.org/10.1016/j.pt.2012.04.005 PMID: 22633470

- World Health Organization. Onchocerciasis fact sheet. Geneva: WHO. Available at:https://www.who. int/news-room/fact-sheets/detail/onchocerciasis (Accessed: February 12, 2022).
- Centers for Disease Control and Prevention. Onchocerciasis—Biology. Available from: https://www. cdc.gov/parasites/onchocerciasis/biology.html (Accessed: February 12, 2022).
- Centers for Disease Control and Prevention. Neglected Tropical Diseases—The Burden of Onchocerciasis. Available from: https://www.cdc.gov/globalhealth/ntd/diseases/oncho_burden.html (Accessed: February 12, 2022).
- Tekle AH, Elhassan E, Isiyaku S, Amazigo UV, Bush S, Noma M, et al. Impact of long-term treatment of onchocerciasis with ivermectin in Kaduna State, Nigeria: first evidence of the potential for elimination in the operational area of the African Programme for Onchocerciasis Control. Parasit Vectors. 2012; 5:28. https://doi.org/10.1186/1756-3305-5-28 PMID: 22313631
- Lakwo T, Garms R, Wamani J, Tukahebwa EM, Byamukama E, Onapa AW, et al. Interruption of the transmission of Onchocerca volvulus in the Kashoya-Kitomi focus, western Uganda by long-term ivermectin treatment and elimination of the vector *Simulium neavei* by larviciding. Acta Trop. 2017; 167:128–136. https://doi.org/10.1016/j.actatropica.2016.12.029 PMID: 28034767
- Higazi TB, Zarroug IM, Mohamed HA, Elmubark WA, Deran TC, Aziz N, et al. Interruption of Onchocerca volvulus transmission in the Abu Hamed focus, Sudan. Am J Trop Med Hyg. 2013; 89(1):51–7. https://doi.org/10.4269/ajtmh.13-0112 PMID: 23690554
- Mushi V. Factors associated with persistence of Onchocerciasis transmission after two decades of Community Directed Treatment with Ivermectin in Ulanga District Council. Masters Thesis. 2018.
- Eisenbarth A, Achukwi MD, Renz A. Ongoing Transmission of Onchocerca volvulus after 25 Years of Annual Ivermectin Mass Treatments in the Vina du Nord River Valley, in North Cameroon. PLoS Negl Trop Dis. 2016; 10(2):e0004392. https://doi.org/10.1371/journal.pntd.0004392 PMID: 26926855
- Bhwana D, Mmbando BP, Dekker MC, Mnacho M, Kakorozya A, Matuja W, et al. Clinical presentation of epilepsy in six villages in an onchocerciasis endemic area in Mahenge, Tanzania. Epileptic Disord. 2019; 21(5):425–435. https://doi.org/10.1684/epd.2019.1093 PMID: 31649005
- Mmbando BP, Suykerbuyk P, Mnacho M, Kakorozya A, Matuja W, Hendy A, et al. High prevalence of epilepsy in two rural onchocerciasis endemic villages in the Mahenge area, Tanzania, after 20 years of community directed treatment with ivermectin. Infect Dis Poverty. 2018; 7(1):64. <u>https://doi.org/10. 1186/s40249-018-0450-3 PMID: 29921319</u>
- König R, Nassri A, Meindl M, Matuja W, Kidunda AR, Siegmund V, et al. The role of Onchocerca volvulus in the development of epilepsy in a rural area of Tanzania. Parasitology. 2010; 137(10):1559–68. https://doi.org/10.1017/S0031182010000338 PMID: 20388236
- Colebunders R, Njamnshi AK, van Oijen M, Mukendi D, Kashama JM, Mandro M, et al. Onchocerciasisassociated epilepsy: From recent epidemiological and clinical findings to policy implications. Epilepsia Open. 2017; 2(2):145–152. https://doi.org/10.1002/epi4.12054 PMID: 29588943
- Mandro M, Suykerbuyk P, Tepage F, Rossy D, Ngave F, Hasan MN, et al. Onchocerca volvulus as a risk factor for developing epilepsy in onchocerciasis endemic regions in the Democratic Republic of Congo: a case control study. Infect Dis Poverty. 2018; 7(1):79. https://doi.org/10.1186/s40249-018-0465-9 PMID: 30293530
- Dusabimana A, Bhwana D, Mandro M, Mmbando BP, Siewe Fodjo JN, Colebunders R. OV16 Seroprevalence among Persons with Epilepsy in Onchocerciasis Endemic Regions: A Multi-Country Study. P,athogens. 2020; 9(10):847. https://doi.org/10.3390/pathogens9100847 PMID: 33081184
- Van Cutsem G, Siewe Fodjo JN, Dekker MCJ, Amaral LJ, Njamnshi AK, Colebunders R. Case definitions for onchocerciasis-associated epilepsy and nodding syndrome: A focused review. Seizure. 2023; 107:132–135. https://doi.org/10.1016/j.seizure.2023.03.024 PMID: 37023626
- Mwaiko GL, Mtoi RS, Mkufya AR. Onchocerciasis prevalence in Tanzania. Cent Afr J Med. 1990; 36 (4):94–6. PMID: 2225028
- Mushi V. Implementation Challenges of Community Directed Treatment with Ivermectin Programme for Control of Onchocerciasis in Ulanga, Tanzania. East African Heal Res J. 2021; 5(2):123–8.
- Rwiza HT, Kilonzo GP, Haule J, Matuja WB, Mteza I, Mbena P et al. Prevalence and incidence of epilepsy in Ulanga, a rural Tanzanian district: a community-based study. Epilepsia. 1992; 33(6):1051–6. https://doi.org/10.1111/j.1528-1157.1992.tb01758.x PMID: 1464263.
- National Bureau of Statistics. The United Republic of Tanzania 2012 population and housing census distributed by administrative Areas. Ministry of Finance. 2013.
- Diagana M, Preux PM, Tuillas M, Ould Hamady A, Druet-Cabanac M. Dépistage de l'épilepsie en zones tropicales: validation d'un questionnaire en Mauritanie [Dépistage de l'épilepsie en zones tropicales: validation d'un questionnaire en Mauritanie]. Bull Soc Pathol Exot. 2006; 99(2):103–7.

- 22. Fisher RS, Cross JH, D'Souza C, French JA, Haut SR, Higurashi N, et al. Instruction manual for the ILAE 2017 operational classification of seizure types. *Epilepsia*. 2017; 58(4):531–542. https://doi.org/ 10.1111/epi.13671 PMID: 28276064
- Bhwana D, Mmbando BP, Dusabimana A, Mhina A, Challe DP, Fodjo JNS, et al. Ivermectin treatment response in two rural villages with a high prevalence of onchocerciasis and epilepsy, Mahenge Tanzania. Afr Health Sci. 2022 Sep; 22(3):607–16. https://doi.org/10.4314/ahs.v22i3.65 PMID: 36910370
- Mshana MI, Silvestri V, Mushi V, Bonaventura WM, Tarimo D, Ngasala B, et al. Burden and factors associated with onchocerciasis transmission among school-aged children after more than 20 years of Community Directed Treatment with Ivermectin in Ulanga district, Tanzania: A school-based cross-sectional study. PLOS Glob Public Health. 2023; 3(5):e0001919. https://doi.org/10.1371/journal.pgph. 0001919 PMID: 37172010
- 25. Bhwana D, Amaral L-J, Mhina A, Matine P, Francis F, Siewe Fodjo JN, et al. Impact of a bi-annual community-directed treatment with ivermectin programme on the incidence of epilepsy in an onchocerciasis-endemic area of Mahenge, Tanzania: A population-based prospective study. PLoS Negl Trop Dis. 2023; 17(6):e0011178. https://doi.org/10.1371/journal.pntd.0011178 PMID: 37379350
- Mazzoti L. Presencia de microfilariasis de Onchocerca volvulus en al liquids cefalorraquides de enfrmos tratados con hetrazan. Rev Inst Salubr Enferm Trop Mex. 1959; 19:1–5. Available from: https://cir.nii.ac.jp/crid/1573668924489270656
- Hotterbeekx A, Lammens M, Idro R, Akun PR, Lukande R, Akena G, et al. Neuroinflammation and Not Tauopathy Is a Predominant Pathological Signature of Nodding Syndrome. J Neuropathol Exp Neurol. 2019; 78(11):1049–1058. https://doi.org/10.1093/jnen/nlz090 PMID: 31553445
- Hotterbeekx A, Vieri MK, Ramberger M, Jozefzoon-Aghai A, Mandro M, Tepage F et al. No Evidence for the Involvement of Leiomodin-1 Antibodies in the Pathogenesis of Onchocerciasis-Associated Epilepsy. Pathogens. 2021; 10(7):845. https://doi.org/10.3390/pathogens10070845 PMID: 34357995
- Nicholas A. Unlocking the hidden burden of epilepsy in Africa: Understanding the challenges and harnessing opportunities for improved care. Health Sci Rep. 2023; 6(4):e1220. <u>https://doi.org/10.1002/hsr2.1220 PMID: 37081998</u>
- Owolabi LF, Adamu B, Jibo AM, et al. Prevalence of active epilepsy, lifetime epilepsy prevalence, and burden of epilepsy in Sub-Saharan Africa from meta-analysis of door-to-door population-based surveys. *Epilepsy Behav.* 2020; 103(Pt A):106846. <u>https://doi.org/10.1016/j.yebeh.2019.106846</u> PMID: 31941583
- Ackermann S, Le Roux S, Wilmshurst JM. Epidemiology of children with epilepsy at a tertiary referral centre in South Africa. *Seizure*. 2019; 70:82–89. https://doi.org/10.1016/j.seizure.2019.06.018 PMID: 31323565
- **32.** Mushi V. Simulium surveillance and control in Mahenge, Tanzania: time to think bigger and utilise drone-based remote sensing technology. Bull Natl Res Cent 2023 471. 2023; 47(1):1–4. Available from: https://bnrc.springeropen.com/articles/10.1186/s42269-023-01016-x
- Paul A, Adeloye D, George-Carey R, Kolčić I, Grant L, Chan KY. An estimate of the prevalence of epilepsy in Sub-Saharan Africa: A systematic analysis. J Glob Health. 2012; 2(2):020405. <u>https://doi.org/10.7189/jogh.02.020405</u> PMID: 23289080
- 34. Gumisiriza N, Kaiser C, Asaba G, Onen H, Mubiru F, Kisembo D, et al. Changes in epilepsy burden after onchocerciasis elimination in a hyperendemic focus of western Uganda: a comparison of two population-based, cross-sectional studies. Lancet Infect Dis. 2020; 20(11):1315–1323. <u>https://doi.org/10. 1016/S1473-3099(20)30122-5 PMID: 32598869</u>
- Jada SR, Amaral LJ, Lakwo T, Carter JY, Rovarini J, Bol YY, et al. Effect of onchocerciasis elimination measures on the incidence of epilepsy in Maridi, South Sudan: a 3-year longitudinal, prospective, population-based study. Lancet Glob Heal. 2023; 11(8):e1260–e1268. <u>https://doi.org/10.1016/S2214-109X</u> (23)00248-6 PMID: 37474232
- Crump A, Ömura S. Ivermectin, 'wonder drug' from Japan: the human use perspective. Proc Jpn Acad Ser B Phys Biol Sci. 2011; 87(2):13–28. https://doi.org/10.2183/pjab.87.13 PMID: 21321478
- Ngarka L, Siewe Fodjo JN, Ambomatei C, Njamnshi WY, Taryunyu Njamnshi JN, Nfor LN, et al. Epidemiology of epilepsy and relationship with onchocerciasis prevalence in villages of the Ntui Health District of Cameroon. Epilepsy Behav. 2023; 142:109184. <u>https://doi.org/10.1016/j.yebeh.2023.109184</u> PMID: 36972641
- Chesnais CB, Bizet C, Campillo JT, Njamnshi WY, Bopda J, Nwane P, Pion SD, Njamnshi AK, Boussinesq M. A Second Population-Based Cohort Study in Cameroon Confirms the Temporal Relationship Between Onchocerciasis and Epilepsy. Open Forum Infect Dis. 2020; 7(6):ofaa206. https://doi.org/10. 1093/ofid/ofaa206 PMID: 32587878

- Hotterbeekx A, Namale Ssonko V, Oyet W, Lakwo T, Idro R. Neurological manifestations in Onchocerca volvulus infection: A review. Brain Res Bull. 2019; 145:39–44. <u>https://doi.org/10.1016/j.brainresbull.2018.08.024</u> PMID: 30458251
- 40. Colebunders R, Njamnshi AK, Menon S, Newton CR, Hotterbeekx A, Preux PM, et al. Onchocerca volvulus and epilepsy: A comprehensive review using the Bradford Hill criteria for causation. PLoS Negl Trop Dis. 2021; 15(1):e0008965. https://doi.org/10.1371/journal.pntd.0008965 PMID: 33411705
- Bhwana D, Das L, Siewe Fodjo JN, Francis F, Challe DP, Makunde HW, et al. A peer support group intervention to decrease epilepsy-related stigma in an onchocerciasis-endemic area in Mahenge, Tanzania: A pilot study. Epilepsy Behav. 2021; 124:108372. <u>https://doi.org/10.1016/j.yebeh.2021.108372</u> PMID: 34757262